Author’s response to reviews

Title: Characterisation of circulating, clinical Influenza isolates from Bali, Indonesia: preliminary report from the BaliMEI project.

Authors:
Wiku Adisasmito (wiku.adisasmito@gmail.com)
Sri Budayanti (nyomansribudayanti@gmail.com)
Dewi Aisyah (dewi.aisyah.14@ucl.ac.uk)
Tiziano Gallo Cassarino (t.cassarino@ucl.ac.uk)
James Rudge (james.rudge@lshtm.ac.uk)
Simon Watson (sw10@sanger.ac.uk)
Zisis Kozlakidis (z.kozlakidis@ucl.ac.uk)
Gavin Smith (gavin.smith@duke-nus.edu.sg)
Richard Coker (richard.coker@lshtm.ac.uk)

Version: 4 Date: 27 Jul 2017

Author’s response to reviews:

Dear Editor,

Thank you for your kind words and advice throughout.
You have requested to replace the expression in lines 225-228 that "sequence data are only available on reasonable request", with one stating that "sequence data can be taken from the "supplementary materials 2"".
This has now been addressed in the attached manuscript.

Sincerely yours,

Zisis Kozlakidis